Author’s response to reviews

Title: High efficacy of a dimeticone-based pediculicide following a brief application: in vitro assays and randomized controlled investigator-blinded clinical trial

Authors:

Jorg Heukelbach (heukelbach@web.de)
Doerte Wolf (doerte.wolf@cardiosec.de)
John Clark (jclark@vasci.umass.edu)
Hans Dautel (dautel@insectservices.de)
Kristina Roeschmann (k.roeschmann@Pohl-Boskamp.de)

Version: 1 Date: 02 May 2019

Author’s response to reviews:

Dear Megan Barling,

Please find enclosed the revised manuscript.

The following changes/additions have been made:

1. CONSORT checklist has been uploaded, and a statement on the adherence to CONSORT guidelines has been included in the manuscript.

2. Presentation of trial registration has been revised in the abstract.

3. Statement on minor consent has been included in the "Ethics approval and consent to participate" section.

4. Declarations section has been rechecked.

Best regards,

Jorg Heukelbach